Author's response to reviews

Title: Acceptability and adoption of handheld computer data collection for public health research in China.

Authors:

Xia Wan (wanxia.china@gmail.com)
H. Fisher Raymond (hfissherraymond@yahoo.com)
Tiancai Wen (tiancai.wen@gmail.com)
Ding Ding (diding@ucsd.edu)
Qian Wang (749051318@qq.com)
Sanghyuk S Shin (tyreepride@gmail.com)
Gonghuan Yang (yangghuan@vip.sina.com)
Wanxing Chai (wanxing@gmail.com)
Peng Zhang (zhangpmy@gmail.com)
Thomas E Novotny (tnovotny@mail.sdsu.edu)

Version: 4 Date: 20 May 2013

Author's response to reviews:

Dear BMC,
Thanks for the opportunity to again revise our manuscript.

However before acceptance, we ask that you carefully copyedit your manuscript to ensure there are no minor errors. For example, we noted that Table 1 is not referenced in the text. Please correct this along with any other small errors that might be present.

Table 1 is referenced in the text on page 7.

Please also ensure that your revised manuscript conforms to the journal style (http://www.biomedcentral.com/info/ifora/medicine_journals ). It is important that your files are correctly formatted.

We have reviewed the entire manuscript with an eye for journal style. We hope we have caught any problems.

The Authors